Author's response to reviews

Title: GenDrux: A biomedical literature search system to identify gene expression-based drug sensitivity in breast cancer

Authors:

Chiquito Crasto (chiquito.crasto@gmail.com)
Dajie Luo (dluo@stat.wvu.edu)
Feliciano Yu (FYu@peds.uab.edu)
Andres Forero (mate@uab.edu)
Dongquan Chen (dongquan@uab.edu)

Version: 8 Date: 24 April 2011

Author's response to reviews:

Dear Editor,

The revision is attached. Please let me know if further modification is needed.

Dongquan